Author’s response to reviews

Title: Cytokine response to selected MTB antigens in Ghanaian TB patients, before and at 2 weeks of anti-TB therapy is characterized by high expression of IFN-gamma and Granzyme B and inter-individual variation

Authors:

Gloria I Mensah (gmensah@noguchi.ug.edu.gh)
Kennedy K Addo (kaddo@noguchi.ug.edu.gh)
John KA Tetteh (jtteh@noguchi.ug.edu.gh)
Sandra A Sowah (ssowah@noguch.ug.edu.gh)
Thomas Loescher (Loescher@lrz.uni-muenchen.de)
Christof Geldmacher (geldmacher@lrz.uni-muenchen.de)
Dolly Jackson-Sillah (d.jackson-sillah@noguchi.ug.edu.gh)

Version: 4 Date: 21 August 2014

Author’s response to reviews: see over
Response to reviewers

Comment: Table 1 has a number of patient identifiers which could lead to the identification of individuals involved in this study. Please modify or remove details which may allow an individual to be identified. An example would involve the use of age groups ("20-25") instead of specific ages.

Response: To remove potential patient identifiers, We have amended table 1 and numbered the patients numerically. We have also given the age in ranges instead of specific ages.